

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Esaku, Stephen

Article

Does corruption contribute to the rise of the shadow economy? Empirical evidence from Uganda

Cogent Economics & Finance

Provided in Cooperation with: Taylor & Francis Group

Suggested Citation: Esaku, Stephen (2021) : Does corruption contribute to the rise of the shadow economy? Empirical evidence from Uganda, Cogent Economics & Finance, ISSN 2332-2039, Taylor & Francis, Abingdon, Vol. 9, Iss. 1, pp. 1-22, https://doi.org/10.1080/23322039.2021.1932246

This Version is available at: https://hdl.handle.net/10419/270101

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



https://creativecommons.org/licenses/by/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU





Cogent Economics & Finance

ISSN: (Print) (Online) Journal homepage: https://www.tandfonline.com/loi/oaef20

Does corruption contribute to the rise of the shadow economy? Empirical evidence from Uganda

Stephen Esaku |

To cite this article: Stephen Esaku | (2021) Does corruption contribute to the rise of the shadow economy? Empirical evidence from Uganda, Cogent Economics & Finance, 9:1, 1932246, DOI: <u>10.1080/23322039.2021.1932246</u>

To link to this article: <u>https://doi.org/10.1080/23322039.2021.1932246</u>

© 2021 The Author(s). This open access article is distributed under a Creative Commons Attribution (CC-BY) 4.0 license.



0

Published online: 28 May 2021.

Submit your article to this journal 🕝

Article views: 2086



View related articles 🗹

🕨 View Crossmark data 🗹



Citing articles: 5 View citing articles 🗹



Received: 10 February 2021 Accepted: 16 May 2021

*Corresponding author: Stephen Esaku, Department of Economics, Statistics Kyambogo University, Box 810, Soroti, Uganda E-mail: esaku_stephen@yahoo.com

Reviewing editor: Francesco Tajani, Economics and Statistics, University of Bari, ITALY

Additional information is available at the end of the article

GENERAL & APPLIED ECONOMICS | RESEARCH ARTICLE

Does corruption contribute to the rise of the shadow economy? Empirical evidence from Uganda

Stephen Esaku¹*

Abstract: This paper investigates whether corruption has contributed to the rise of the shadow economy in Uganda. Using autoregressive distributed lag bounds testing approach and aranger causality econometric methods we find a positive relationship between corruption and the size of the shadow economy in both the long- and short-run. Additionally, the causality results reveal a bidirectional causal relationship between the shadow economy and corruption, and vice versa. These findings suggest that, for the case of Uganda, an increase in corruption contributes to the rise in the size of the shadow economy and vice versa, all else equal. Given the complementary relationship between corruption and the size of the shadow economy, addressing widespread informality in the country would require; first, reforming the political system to tackle political corruption and go after politicians who use their influence and power to circumvent institutions. Second, carrying out institutional reforms to address political patronage and influence peddling would go a long way into addressing systemic corruption which in turn could help mitigate the spread of informal sector activities. Third, strengthening the enforcement of existing laws to identify and punish culpable public officials who use their offices for private gain would also address the level of informality in the country.

ABOUT THE AUTHOR

Stephen Esaku is a lecturer of Economics in the department of Economics and Statistics Kyambogo University Uganda (Soroti campus). He holds a Phd in Economics specializing in international economics and Development. His research interests are in the areas of international economics (firms in international trade, regional integration, foreign direct investment), and Development economics (focusing on economic growth and development, poverty dynamics, corruption and gender studies) and empirical industrial economics (learning, pricing and market strategies). Stephen has published papers on firm-level investment and Exporting; Export markets and productivity; employment dynamics in Sub-Saharan Africa; Job creation and destruction in an African economy, among others. His teaching experiences spans 8 years.

PUBLIC INTEREST STATEMENT

There is empirical evidence showing the rise and expansion of informal sector activities and corruption in Uganda. Both informality and corruption are detrimental to the growth of a country because they distort the proper functioning of the market mechanism, requiring government's intervention to correct them. Previously, a number of papers have analyzed these variables independently but little is known on how the interaction between these two variables affects each other in Sub-Saharan African context. This paper fills this gap. This paper finds that an increase in corruption increases the size of the shadow economy in both the short- and long-run and a rise in informality results into corruption.

🔆 cogent

economics & finance





 ${\small ©}$ 2021 The Author(s). This open access article is distributed under a Creative Commons Attribution (CC-BY) 4.0 license.

Subjects: African Studies; Introductory Politics; Development Economics; Political Economy; Public Finance; Entrepreneurship

Keywords: Corruption; informal sector; tax evasion and avoidance; country-level studies; shadow economy

GEL classfications: D73; H26; K49; O5; O17

1. Introduction

The existence of corruption¹ and the shadow economy² or informal sector has become a recognized fact of everyday life across the world. These vices seem to be common and expanding, especially in countries where the institutional framework is not sufficient to support the proper functioning of the market mechanism (see Elgin & Oztunali, 2014; De Soto, 1989). To curb their expansion, most countries try to institute measures that make it costly to engage in these activities. Such measures include prosecution and punishment, and in some cases sensitization and ensuring improvements in the provision of public services. Corruption and the shadow economy undermine governments' ability to ensure efficient collection of taxes and provision of public services (Esaku, 2021e). Moreover, both corruption and the shadow economy create economic distortions in the efficient allocation of resources that sustain the livelihoods of the population. Additionally, the shadow economy has also been noted to create more pressure on the formal economy by increasing competition for resources that would otherwise be available for firms that are formally registered and paying taxes (Schneider & Enste, 2000). The above distortions create biases in macroeconomic indicators like consumption expenditure, income level, participation rates of the labor force and unemployment level leading to misrepresentation of actual economic reality in the country (Blackburn et al., 2012; Capasso & Jappelli, 2013; Esaku, 2021e). Consequently, policy makers who rely on data with these distortions may find difficulty in crafting effective monetary and fiscal policies to guide economic growth.

Recent research has attempted to study the shadow economy and corruption in isolation (independent of each other), with a small number of papers explicitly addressing the connection between these variables (see Buehn & Schneider, 2012a; Choi & Thum, 2005; Dreher & Schneider, 2010; Goel & Saunoris, 2014). Theoretical models have been advanced to show how these variables interact. For example, Choi and Thum (2005) develop a theoretical framework to examine the relationship between the shadow economy and corruption in the presence of entrepreneurial heterogeneity. In their model, they show that entrepreneur's incentive to operate underground constrains the ability of the corrupt official to use public power for private gain thereby enhancing the operations of the formal economy. Viewed from this lens, corruption and the shadow economy are negatively correlated (substitutes) as the latter tempers any distortions introduced by corruption. The above authors conclude that the formal economy and the informal economy should be seen as complements instead of substitutes because of the ability of the latter to enhance economic activities in the former. Similarly, Johnson et al. (1997) present a model in which they consider the allocation of labor between the shadow and the formal economy and present evidence to support a positive relationship (complementary relationship) between corruption and the shadow economy. The above authors show that corruption acts like an additional tax to the entrepreneur so that its presence induces more entrepreneurs to operate underground thereby concealing their production and distribution of goods and services from the "eye" of the public officials.

Correspondingly, Johnson et al. (1998) study how bribes act as a driving force in the growth of the shadow economy and find evidence to support a positive relationship between these variables. The above authors show that in economies where the rule of law is weak, the presence of corruption leads to substantial growth of the informal economy. Their study emphasizes that it is not taxes or regulation in itself that is important but the role governments play in controlling corruption. Friedman et al. (2000) reach similar conclusions in which they show that corruption and

the shadow economy are complements. Similar findings are shown by Hibbs and Piculescu (2005) who emphasize that the "grabbing hands" of the corrupt official may also serve as the "helping hands" that induce entrepreneurs, who see the "grabbing hands" as an extra burden, to operate underground.

On the empirical side, some studies have examined the relationship between corruption and the shadow economy. For example, Buehn and Schneider (2012a) investigate the relationship between the show economy and corruption using a sample of 51 countries around the world and show evidence of a positive relationship between the two variables. In their analysis, the high levels of corruption lead to the growth of the informal economy, as entrepreneurs try to circumvent the "extra cost" imposed by corruption. Moreover, the above authors show that less regulation and lower levels of taxation dampen the ability of corrupt officials to ask for bribes as most entrepreneurs prefer to operate in the formal economy to benefit from government incentives that may be available to formal firms. Additionally, Schneider (2007) presents estimates of the shadow economy across 145 countries and shows that the shadow economy increases corruption in low income countries and reduces corruption in high-income countries. Goel and Saunoris (2014) analyze the relationship between the shadow economy and corruption and find evidence to support a positive relationship between the shadow economy and corruption. Conversely, Dreher and Schneider (2010) use data from a cross-section of 98 countries to study the relationship between corruption and the shadow economy. Their empirical analysis shows no robust link between the shadow economy and corruption when indices of perception of corruption are used, except when the index of corruption from the structural model is used. The use of index of corruption shows that the shadow economy and corruption are positively correlated, especially in lowincome countries.

From both the theoretical and empirical literature, there is evidence that the relationship between the shadow economy and corruption is still a contentious issue. Theoretically, the relationship between the shadow economy and corruption could either be a non-reinforcing one (see Choi & Thum, 2005) or a reinforcing one (positive) (see Friedman et al., 2000; Hibbs & Piculescu, 2005; Johnson et al., 1998). This theoretical evidence shows that whether the shadow economy and corruption should be seen as complements or substitutes is still a controversial issue. Furthermore, the empirical examination of the relationship between the shadow economy and corruption has also produced mixed results. For example, the empirical relationship could either be positive (Buehn & Schneider, 2012a) or negative (Dreher et al., 2009; Schneider, 2007), but also it could depend on the income level of the country under investigation (Dreher & Schneider, 2010; Schneider, 2007), and the corruption index used (Dreher & Schneider, 2010).

In line with the above controversy, it seems plausible to suggest that the informal economy and corruption may be linked, especially in countries where corruption thrives unabated. In such countries, entrepreneurs who see taxes paid by the formal sector as excessive may be motivated to conceal their operations to avoid detection. These entrepreneurs may pay small bribes to continue operating underground. However, it is also plausible to view corruption and the shadow economy as substitutes especially where institutions are strong and effective, and where informal businesses frustrate rent-seeking avenues for the corrupt government official thereby reducing the ability of the public official to seek bribes and introduce economic distortions to the economy. Whether the relationship between the informal economy and corruption is positive or negative (complementary or substitution), what is clear is that it varies across economies of the world. This implies that there is no established pattern in the relationship between corruption and the shadow economy, and one cannot draw general conclusions that apply across the economies of the world. This is one limitation in the extant literature. Furthermore, most of the theoretical and empirical work on the relationship between corruption and the shadow economy has focused on cross-country analysis with limited attention on how the two variables interact at the sub-national and

country levels. Additionally, the bulk of the papers have studied the above relationship in the context of developed countries with less focus on less developed settings. Given this, it becomes essential to study the above relationship in a country-level framework.

In this paper, we investigate whether corruption has contributed to the rise of the shadow economy in Uganda. The few studies that have investigated this relationship focused on contexts outside of Africa with none examining this relationship in the context of a low-income country like Uganda. It is plausible to suggest that investigating the above relationship in the context of a low-income country like Uganda could provide fresh insights into this relationship that may help in formulation of policies to control the growth and expansion of the two vices (Mugoda et al., 2020). In this paper, we specifically ask, if this relationship is positive (complementary), what policy framework may be needed to control the rise of the two vices? We make two important contributions to the literature; first, we study the relationship between the shadow economy and corruption in the context that is less studied (low-income country). Empirical investigation of this relationship in the above context provides fresh insights that could be generalized to similar contexts. Second, we study an important relationship that has received little examination over the past decades. Corruption and the shadow economy are two factors that retard economic growth and development across many countries. Analyzing how they are intrinsically related would be important in advancing theory and empirical work on how they can be addressed in low-income countries.

Apart from section 1, section 2 reviews related literature with section 3 presenting the data and descriptive statistics. Section 4 presents the methodology while section 5 reports the findings and discussion. Section 6 concludes this paper.

2. Review of related literature

2.1. Theory and empirical evidence

The shadow economy and corruption remain a fact of life around the world but difficult to measure. Part of the reason for this is that activities in the shadow economy and corrupt practices are hidden because they are done underground. Accurate measures of these activities are hard to come by, whatever available evidence that there is, comes from World Bank surveys and other international organizations. A number of papers have examined the shadow economy and corruption separately from each other, with a limited attempt to analyze how they may interact with each other. There is a possibility that activities in the shadow economy and corrupt practices might be interrelated since they are done underground. Consequently, examining them separately could result into overlooking of the possibility that they may be interrelated, and making measures to control them ineffective (see Goel & Saunoris, 2014). Theoretically, some papers have presented different mechanisms showing how the shadow economy may interact with corruption. For example, Choi and Thum (2005) present a simple model to examine the relationship between the shadow economy and corruption and how their interrelationship affects the formal economy. They show that entrepreneurs' activities in the shadow economy limit the ability of corrupt officials to use their regulatory power for private gain, which would otherwise introduce distortions to the economy. Given the above, the shadow economy alleviates "governmentinduced distortions" leading to increased business activities in the formal economy (Esaku & Watson, 2020b). Thus, the relationship between the shadow economy and corruption is one of substitutes rather than complements, as the shadow economy activities reinforce the formal sector.

The theoretical evidence in Choi and Thum (2005) contradicts that of some studies. For instance, Johnson et al. (1997) examine the shadow economy in transition countries and report evidence of a positive relationship between the shadow economy and corruption. Similar

conclusions are reached in Johnson et al. (1998). In their model, corruption can be seen as imposing an "extra cost" to the entrepreneurs who want to formalize their businesses, thus provides the incentive for these entrepreneurs to conceal their operations from the radar of regulators. Consequently, any increase in the level of corruption leads to a rise in the shadow economy so that they can be taken as complements. Similarly, in Hindricks et al. (1999) model of the shadow economy, the corrupt tax official can reduce the entrepreneur's tax liability in exchange for a bribe, consequently increasing the level of shadow activities in the economy. This is because these entrepreneurs have the incentive to take advantage of profitable opportunities outside of the formal sector. They show that the entrepreneur can collude with the corrupt tax official so he/she (corrupt tax official) under-reports the entrepreneur's tax liability in return for a bribe. Their study provides theoretical evidence that supports a positive relationship between corruption and the shadow economy. In the same vein, Hibbs and Piculescu (2005) emphasize that the "grabbing hands" of the corrupt official could also potentially become the "helping hands" for entrepreneurs who see the "grabbing hands" as an "extra burden" to their business hence inducing these entrepreneurs to conceal their operations from the regulators. In this case, the shadow economy and corruption could be taken as reinforcing each other rather than tempering each other.

Some other studies have examined the empirical relationship between the shadow economy and corruption and find evidence suggesting that their relationship could be either negative or positive. And in some cases it depends on the income level of the economies under investigation. The extent to which the shadow economy and corruption are either substitutes or complements could possibly vary between low and high-income countries and the type of corruption index used (see Dreher & Schneider, 2010). Thus, Buehn and Schneider (2012a) analyze the link between corruption and the show economy in a sample of 51 countries and find evidence that supports a positive relationship between corruption and the shadow economy. They show that high levels of corruption also create high levels of shadow activities as businesses try to overcome the "extra burden" imposed by corruption. In this sense, less regulation and lower levels of taxation reduce the ability of corrupt officials to extort money from entrepreneurs since these businesses can conceal their operations from government regulators. Furthermore, Friedman et al. (2000) find evidence showing that entrepreneurs operate in the shadow economy because of discretion in the application of regulations and the existence of corruption. They emphasize that high levels of corruption and the existence of a weak institutional framework induce entrepreneurs to abandon the formal sector for the shadow economy. Similarly, Goel and Saunoris (2014) investigate the relationship between corruption and the shadow economy and find evidence in support of a positive relationship.

Conversely, Dreher and Schneider (2010) use data from a cross-section of 98 countries to study the relationship between corruption and the shadow economy. They challenge results that support a positive (complementary) relationship between the shadow economy and corruption. They provide evidence of a positive relationship only when the index of corruption from the structural model is used. However, when perception-based index is used, they find no robust relationship between the shadow economy and corruption suggesting that they substitute for one another. The use of index of corruption shows that the shadow economy and corruption are complements, especially in low-income countries. Correspondingly, Dreher et al. (2009) analyze the relationship between corruption and the shadow economy in samples of 78 and 135 countries and find evidence of a negative relationship. Their empirical evidence shows that the informal economy can alleviate government-induced economic distortions arising from corrupt practices so that the shadow economy and corruption could be considered substitutes rather than complements. The above conjecture could possibly mirror a picture of a developed economy where soliciting for bribes may result in prosecution because of a strong institutional framework in those countries. However, this might not be the case for low-income countries given the weak nature of institutions in these countries.

While one could expect some anecdotal evidence of the relationship between the shadow economy and corruption in other contexts, a formal evaluation of this relationship is required in a less developed setting to provide further evidence. This paper attempts to provide evidence for the less developed context, Uganda. There is a possibility that the two activities (corruption and informality) could be related and need to be examined simultaneously in different contexts as evidence suggests that the geography of corruption and informal economy matters (Dreher & Schneider, 2010). Furthermore, the relationship between the shadow economy and corruption has rarely been studied in a single-country context (to the best of our knowledge), especially in a low-income country like Uganda. This could possibly be the first attempt to unravel the mystery in the relationship between the shadow economy and empirical evidence do not seem to provide a definitive answer to the relationship between corruption and the shadow economy, as emphasized by Buehn and Schneider (2012b).

3. The data and descriptive statistics

In this section, we present the data and descriptive statistics of the main variable under this study. This paper uses annual time series data obtained from various internationally recognized sources, from 1984 to 2008. We report the summary statistics for the data in Table 1. The data for our outcome variable, the shadow economy, are drawn from Elgin and Oztunali (2012), who provide the estimates of the size of shadow economy based on two-sector dynamic model. Next, the first explanatory variable, corruption, is from the International Country Risk Guide (International Country Risk Guide (ICGR) Researchers Dataset, 2017) that measures corruption for an extended period of time. This corruption index ranges from 0, denoting the highest level of corruption, to 6, representing no corruption (Dreher et al., 2009). This implies that lower scores indicate the high likelihood of government officials demanding special and illicit payments, in form of bribes, likely throughout lower levels of government (Knack & Keefer, 1995). For robustness checks, we also use perception-based indices of corruption, in an alternative specification, drawn from Transparency International (TI). The subsequent control variables include gross domestic product (GDP) growth annual percent, general government final consumption expenditure as (% of GDP), polity2 and institutional quality. The first two variables, that is, gross domestic product (GDP) growth annual percent, and general government final consumption expenditure as (% of GDP), are all drawn from Africa development indicators of the World Bank ("World bank," 2014). Polity2, our proxy measure of democracy is from polity IV project (Marshall & Gur, 2018), while institutional quality (inst.qual) is simple average of three indices: (i) rule of law, (ii) democratic accountability, and (iii) bureaucratic quality. All these three variables are from ICGR published by Political Risk Services (PRS), and widely used indices to measure the quality of institution (International Country Risk Guide (ICGR) Researchers Dataset, 2017).

In Table 1, we can observe that the average values of the key variables are shadow economy (shadow E) is 44.22, corruption (corr) 2.26, institutional quality (inst.qual.) is 2.38, growth is 2.93, government spending (GOV/GDP) is 11.9, polity2 and corruption-based indices (Corp), are -3.92 and 24.47, respectively. Apart from summary statistics, we also include the correlation matrix in Table 2. From Table 2, we can observe that there is a positive correlation between the size of the shadow economy and measure of corruption. However, one can argue that positive correlation does not necessarily imply causation. We should embark on a formal analysis of the relationship between corruption and the size of the shadow economy.

4. Methodology

In this section we present the methodology used to examine the relationship between the shadow economy and corruption. We conduct empirical analysis in four basic steps; first, we specify our

| Table 1. Sumr | Table 1. Summary statistics of the key variables | | | | | | | | |
|---------------|--|-------|----------|-------|-------|--|--|--|--|
| | Obs | Mean | Std.Dev. | Min | Max | Data source | | | |
| Shadow E. | 25 | 48.22 | 5.01 | 39.38 | 54.80 | Elgin and Oztunali (2012) | | | |
| Corruption | 25 | 2.26 | 0.64 | 1.00 | 3.00 | International Country Risk Guide (International Country Risk Guide (ICGR) Researchers Dataset, 2017) | | | |
| Inst.Quality | 25 | 2.38 | 0.56 | 1.33 | 3.00 | International Country Risk Guide (International Country Risk Guide (ICGR) Researchers Dataset, 2017) | | | |
| Growth | 25 | 2.93 | 3.34 | -6.48 | 8.11 | "Africa development indicators, World Bank" (2014) | | | |
| GOV/GDP | 25 | 11.90 | 2.77 | 6.97 | 16.79 | "Africa development indicators, World Bank" (2014) | | | |
| Polity2 | 25 | -3.92 | 2.61 | -7.00 | 3.00 | Center for Systemic Peace (CSP), Polity V (Marshall et al. 2018) | | | |
| Corp | 13 | 24.47 | 2.77 | 19.00 | 28.00 | Transparency International (2019) | | | |

Source: Author's calculation.

| Table 2. Co | orrelation m | atrix of the | core variabl | es | | | |
|-------------|--------------|--------------|--------------|--------|---------|---------|------|
| | Shadow | Corr | Inst. Qual | Growth | GOV/GDP | Polity2 | Corp |
| Shadow | 1.000 | | | | | | |
| Corr | 0.485 | 1.000 | | | | | |
| Inst.Qual | -0.381 | 0.366 | 1.000 | | | | |
| Growth | -0.357 | 0.308 | 0.475 | 1.000 | | | |
| GOV/GDP | -0.020 | -0.392 | -0.172 | -0.246 | 1.000 | | |
| Polity2 | -0.506 | -0.192 | 0.466 | 0.464 | -0.291 | 1.000 | |
| Corp | -0.187 | 0.286 | 0.520 | 0.344 | -0.600 | 0.510 | 1.00 |

Source: Author's calculation.

estimation model. We specify our estimation techniques in step two and determine the cointegration relationship (whether variables are I(0) or I(1)) in step three. In step four, we conduct the empirical analysis and perform diagnostic tests to confirm the robustness of the results.

4.1. Model specification

The empirical model that we use for testing the relationship between corruption and the shadow economy is in line with the literature reviewed above. Consequently, we adapt the framework of Dreher and Schneider (2010). Since the interest of this paper is to investigate whether corruption contributes to the rise in the shadow economy (and not whether the shadow economy promotes corrupt activities), we express our empirical model as follows:

$$se_i = \beta_0 + \beta_1 cor_i + \beta_2 Z_i + \mu_i \tag{1}$$

Where se_i denotes the shadow economy, cor_i is the measure of corruption, cop_i is corruption perception measure, Z_i is the vector of variables that control for fiscal, macroeconomic and institutional environment. These variables include GDP growth per capita (annual percent), government final consumption expenditure as a ratio of GDP, measure of democracy (polity2) and institutional quality. Additionally, μ_i denotes the error term. The rationale for the inclusion of the above variables in eq. (1) is well documented in the extant literature on the shadow economy and corruption (see Buehn &

Schneider, 2012a; Dreher et al., 2009; Dreher & Schneider, 2010; Esaku, 2021c; Johnson et al., 1997, 1998). For example, empirical evidence has shown that higher taxes have a positive and statistically significant impact on the shadow economy (Dreher et al. 2009). However, data on the tax burden is scarce so analysts have used government consumption expenditure to proxy for the tax burden since higher government expenditure implies an increase in taxes to fund public goods and services like security. An increase in taxes provides incentives for entrepreneurs to operate underground to avoid paying taxes. Furthermore, when entrepreneurs go underground this means that governments cannot raise adequate revenue to provide public goods and services. Funding for infrastructure projects, law and order, security, and education among others, will be affected which leads to low level of economic growth and development (Esaku, 2021b). Consequently, one can argue that the rise of the size of the shadow economy affects the level of development in a country (Esaku, 2021b). In sum, the above variables broadly reflect the level of institutional framework and macroeconomic conditions that likely determine the size of the shadow economy.

4.2. Estimation technique

This paper follows three steps for estimating the relationship between corruption and the shadow economy. First, we conduct unit root tests to determine whether the main variables are integrated of order zero and or order one, that is, I(0)s or I(1)s. Second, we conduct tests of the presence of long-run relationships (cointegration tests) between the variables, and third, we implement the empirical tests and conduct model diagnostics. Thus, we follow the autoregressive distributed lag (ARDL) bounds testing approach to cointegration of Pesaran et al. (2001), which has a number of merits over traditional cointegration techniques. This approach is robust for determining long-run and short-run relationships in the presence of small sample size (Tang, 2004). Further, this approach can be carried out regardless of whether the variables are I(0)s and or I(1)s (Pesaran et al., 2001). To analyze the relationship between the shadow economy and corruption, this study uses the autoregressive distributed lag (ARDL) bounds testing method to cointegration in line with Pesaran et al. (2001). Thus, the ARDL specification can be expressed as:

$$\Delta se = \beta_0 + \sum_{i=1}^n \beta_{1i} \Delta se_{t-i} + \sum_{i=0}^n \beta_{2i} \Delta corr_{t-i} + \sum_{i=0}^n \beta_{3i} \Delta iq_{t-i} + \sum_{i=0}^n \beta_{4i} \Delta growth_{t-i} + \sum_{i=0}^n \beta_{5i} \Delta gov/gdp_{t-i} + \sum_{i=0}^n \beta_{5i} \Delta gov/gdp_{t-i} + \gamma_1 se_{t-i} + \gamma_2 corr_{t-1} + \gamma_3 iq_{t-1} + \gamma_4 growth_{t-i} + \gamma_5 gov/gdp_{t-1} + \gamma_6 pol_{t-i} + \varepsilon_t$$
(3)

Where β_0 is the constant term while β_1, \ldots, β_6 denote the short-run coefficients, while $\gamma_1, \ldots, \gamma_6$, denote long-run coefficients, and finally, ε_t is the error term.

The ARDL bounds testing technique in this paper follows two steps. First, we test for cointegration among variables with the purpose of determining whether there exists a long-run relationship. The existence of the long-run relationship among variables is tested using the F-statistic and t-statistic. From the calculated F-statistic, the null hypothesis of no cointegration, that is, $(H_0 = \alpha_1 = \alpha_2 = \alpha_3 = \alpha_4 = \alpha_5 = \alpha_6 = 0)$, against the alternative hypothesis of cointegration, that, $(H_1 \neq \alpha_1 \neq \alpha_2 \neq \alpha_3 \neq \alpha_4 \neq \alpha_5 \neq \alpha_6 \neq 0)$. The results of the calculated F-statistic are compared with the critical values in Pesaran et al. (2001). The decision rule is that, if the calculated values of the F-statistic exceed the upper critical bound values, the H_0 (the null hypothesis) is rejected and vice versa. On the other hand, if the calculated values of F-statistic fall within the bounds, then the test results are inconclusive. In the second step, we employ ARDL bounds testing procedure to test for relationships. During this step, the optimal lag length for the ARDL model is chosen using appropriate lag selection criteria based on the Schwartz-Bayesian criterion (SBC).

If the results of the cointegration test on equation (3) indicate cointegration among variables, then we can proceed to express the error correction model (ECM) as:

$$\Delta se_{t} = \beta_{0} + \sum_{i=1}^{n} \beta_{1i} \Delta se_{t-i} + \sum_{i=0}^{n} \beta_{2i} \Delta corr_{t-i} + \sum_{i=0}^{n} \beta_{3i} \Delta iq_{t-i} + \sum_{i=0}^{n} \beta_{4i} \Delta growth_{t-i} + \sum_{i=0}^{n} \beta_{5i} \Delta gov/gdp_{t-i} + \sum_{i=0}^{n} \beta_{6i} \Delta pol2_{t-i} + \lambda ECT_{t-1} + \varepsilon_{t}$$

$$(4)$$

where ECT represents the error correction term, while λ denotes the coefficient of the error correction term which captures long-run adjustments and ε_t being the residual error term. The size and sign of the error correction term determines the economic significance of the estimates, and speed of adjustment. This implies that the coefficient of the error correction term λ should be negative, less than one and statistically significant (Enders, 2004).

5. Results and discussion

5.1. Stationarity tests

In this section, we report the findings of the pre-requisite tests before conducting empirical analysis. Before conducting the analysis, we first tested the variables for the existence of unit roots to determine whether they are integrated of order zero, I(0), or order one, I(1) to enable a valid estimation process. This paper uses two varying tests to conduct unit root tests. Accordingly, we use Augmented-Dickey-Fuller test (ADF), and Phillip-Perron (PP) test. The results of these tests are presented in Table 3. The test results in Table 3 show that the variables are stationary after first differencing. Moreover, our results vary depending on the type of test employed. This study then proceeded to implement ARDL bounds testing after the stationarity test was conducted. In Table 4 we report the results of the ARDL bounds test for cointegration to establish whether there exists a long-run relationship. In Table 4 we can note that the calculated F-statistic is higher than the asymptotic critical value bounds. This leads to the rejection of the null hypothesis of no cointegration and the conclusion that the variables are cointegrated. Once the cointegration test was carried out, this study proceeded to estimate the long- and short-run coefficients for the model. Consequently, this study determined the optimal lag length using the Schwartz information criterion (SIC), which selected ARDL (1,2,2,0,0,2) for model (1), and ARDL (1,0,0,0,0,0) for model (2). Models (1) and (2) denote control of corruption, and perception-based indices of corruption respectively.

5.2. The long-run relationship between the shadow economy and corruption

Having determined the optimal lag length for the respective models we then conducted the empirical estimation of the long- and short-run relationship between corruption and the shadow economy. We report the results of the empirical analysis of the long-run relationship between the shadow economy and corruption, using ARDL, DOLS (dynamic ordinary least squares) and FMOLS (fully modified ordinary least squares) in Table 5, columns 2, 4 and 6, respectively. In Table 5, column 2, this study establishes a positive and statistically significant long-run relationship between the shadow economy and corruption. Thus, the coefficient on corruption is positive and statistically significant at 1% level. The above findings are not surprising given that Uganda is a low-income country with institutions that exist but unable to enforce measures aimed at curbing corruption. These results seem to bode well with the findings of Buehn and Schneider (2012a). According to Buehn and Schneider (2012a), the level of corruption and the size of the shadow economy exhibit a positive relationship. Additionally, Dreher and Schneider (2010) study suggests that the relationship between corruption and the shadow economy exhibits a stronger positive relationship in low income countries than high income countries. This appears to be the case with our results. Given than Uganda is a low income country, these results are consistent with the findings in Dreher and Schneider (2010). Similarly, Hindricks et al. (1999) suggest that the presence of corruption induces the corrupt tax official to collude with the entrepreneur for his/her private gain, by reducing the tax liability of the entrepreneur in exchange for a bribe. The reduction in the entrepreneur's tax liability is a strong incentive for colluding entrepreneurs to continue operating in the informal economy. This might seem to be the case for Uganda.

| Table 3. Res | ults of the uni | t root test | | | | | | | | |
|--------------------------------------|--|---|-------------------|--------------------|--------------------|--------------------|--------------------|---------------------|---------------------|-----------------|
| Variable | AI | ugmented-Dic | key-Fuller (AD |)F) | | Phillip-Pe | erron (PP) | | Structurc | ıl breaks |
| | In le | svels | In Fir | st Diff | In le | ivels | In Fin | st Diff | In levels | First Dif |
| | T&C | Const | T&C | Const | T&C | Const | T&C | Const | | |
| Shadow E | -2.305 | -1.246 | -3.547* | -3.187** | -2.949 | 2.105 | -2.655 | -2.641* | -0.973 | -9.552 |
| Corruption | -3.745** | -2.912* | 1 | I | -2.519 | -2.243 | -2.721** | -3.439** | -11.389** | -7.430** |
| Ins.quality | -3.031 | -1.295 | -4.003** | -3.752** | -1.573 | -1.893 | -3.197 | -3.142** | -13.867** | -7.848** |
| Growth | -3.154 | -2.749* | -6.671*** | -6.584*** | -3.046 | -2.576 | -7.377** | -6.865*** | -4.549** | -7.460** |
| Gov/gdp | -1.792 | -1.505 | -3.583* | -5.444*** | -1.792 | -1.505 | -5.253*** | -5.350*** | -3.880 | -7.427** |
| Polity2 | -4.975** | -2.996** | 1 | -3.678** | -5.809*** | -3.053** | I | -3.636** | -4.629** | -6.430** |
| Corp | -2.412 | -1.845 | -4.175** | -3.827** | -1.780 | -1.748 | -4.150** | -3.787** | -3.572 | -5.905** |
| Source: Author's is 1988, and cor | s calculations. *, **, ruption perceptio. | ,***; denote signifi n (Corp) is 2003. | icance at 10%, 5% | 6 and 1% levels. B | reak dates for Sho | ıdow Ε is 1994; cι | orruption is 1995; | inst.quality and pc | olity2 is1992; grow | /th and gov/gdp |

| dp6// |
|--------------------|
| vog þi |
| th an |
| grow |
| 1992; |
| y2 is: |
| l polit |
| y anc |
| quali |
| ; inst. |
| 1995 |
| ion is |
| irrupt |
| 94; cc |
| is 19 |
| dow E |
| r Sha |
| tes fo |
| ak da |
| s. Bre |
| level |
| nd 1% |
| 5% ar |
| .0%, |
| e at 1 |
| ficano |
| signił 2003. |
| enote rp) is |
| ***; dı n (Coi |
| s. *,**, eptio |
| ation: 1 perc |
| alcul |
| hor's (l corri |
| e: Autl 8, and |
| urce 198(|

| Table 4. Re | Table 4. Results of the ARDL Bounds test | | | | | | | |
|-------------|--|---------------|-------------------------------------|---------|-------------------------------------|------------------------------|--|--|
| Model | ARDL | F-statistic | | | Diagnostics | | | |
| | | | X ² (nor | mality) | X ² (heteroscedasticity) | X ² (correlation) | | |
| Eq. (1) | (1,2,2,0,0,2) | 31.658 | 0.9 | 938 | 0.609 | 0.233 | | |
| Eq. (2) | (1,0,0,0,0,0) | 29.678 | 0.305 | | 0.518 | 0.634 | | |
| | | , | Asymptotic Critical values (T = 25) | | | | | |
| | Lowe | er Bound I(0) | | | Upper Bound I(1) | | | |
| 10% | | 2.26 | | 3.35 | | | | |
| 5% | | 2.62 | | | 3.79 | | | |
| 2.5% | | 2.96 | | | 4.18 | | | |
| 1% | | 3.41 | | | 4.68 | | | |

Source: Author's calculation.

Moreover, these findings are in line with the results of Goel and Saunoris (2014) who show that corruption and the size of the shadow economy are positively correlated.

The implication of these findings is that corruption seems to be the main driver of informality in the long-run, in Uganda. As pointed out by Johnson et al. (1998), one can expect a rise in the level of corruption to increase informality in countries where the institutional framework is weak. This is because corruption could be viewed by the entrepreneurs, as an "additional cost" to be incurred so as to join the formal sector. This "additional cost" induces entrepreneurs to operate underground to avoid regulators, which leads to the increase in informality. Given that corruption significantly increases the size of the shadow economy, addressing it would require; first, reforming the political system to have the political will to tackle political corruption and go after politicians who use their influence and power to circumvent institutions. Second, institutional reforms to address political patronage and influence peddling would go a long way into addressing systemic corruption which in turn would help mitigate the spread of informal sector activities. Third, strengthening the enforcement of existing laws to identify and punish culpable public officials who use their offices for private gain would also address the level of informality in the country.

We further explain the remaining part of the results in Table 5. We can observe that improvement in institutional framework is important in mitigating the expansion of informal sector activities. This study finds that an improvement in institutional quality significantly reduces the size of the shadow economy in the long-run, all else equal. Specifically, an improvement in institutional quality reduces the size of the shadow economy by 0.733 units, statistically significant at 1% level. This finding indicates the importance of a strong institutional framework in mitigating wide spread informal sector activities. As shown in the literature, the quality of institutions in the country is important in enforcement of the tax laws and regulating business practices (see Elgin & Oztunali, 2014; Esaku, 2021a; Luong et al., 2020). Correspondingly, the empirical results also show the importance of the level of economic growth and development in mitigating increases in informal sector activities. We can observe that the coefficient on growth is negative and statistically significant at 1% level. Specifically, the results show that an increase in economic growth reduces the size of the shadow economy by 0.027 units, statistically significant at 1% level. This is consistent with the literature that shows the crucial role of economic growth in reducing the size of informal sector activities (see Esaku, 2021b; Luong et al., 2020). This implies that an increase in annual growth of GDP significantly mitigates increase in informal sector activities. If improvements in economic growth also translate into the betterment of the poor, then the formal sector can

| Table 5. Long-run rel | ationship be | tween shad | ow economy | and corrupt | tion | | | | |
|-----------------------|----------------------------------|------------|-------------|-------------|-------------|---------|--|--|--|
| Explanatory | Outcome variable: shadow economy | | | | | | | | |
| variables | AR | DL | DO | OLS | FM | OLS | | | |
| | Coefficient | t-stat. | Coefficient | t-stat. | Coefficient | t-stat. | | | |
| Corruption | 0.175*** | 3.416 | 0.175*** | 5.161 | 0.145*** | 4.905 | | | |
| Inst. Quality | -0.733*** | -10.221 | -0.733*** | -13.335 | -0.703*** | -14.442 | | | |
| Growth | -0.027*** | -3.306 | -0.027*** | -4.152 | -0.029*** | -5.087 | | | |
| GOV./GDP | 0.063*** | 4.639 | 0.063*** | 7.443 | 0.067*** | 9.145 | | | |
| Polity2 | 0.026** | 2.826 | 0.027*** | 3.417 | 0.023** | 2.573 | | | |
| Constant | 1.078** | 2.368 | 1.078** | 2.749 | 0.815** | 2.437 | | | |
| R-Squared | 0.998 | | 0.998 | | 0.997 | | | | |
| R-bar-Squared | 0.997 | | 0.997 | | 0.996 | | | | |
| Durbin Watson | 1.912 | | | | | | | | |

Source: Author's calculation. **, ***, indicate statistical significance level at 5% and 1% respectively.

| Table 6. Short-run rela | Table 6. Short-run relationship between shadow economy and corruption | | | | | | | | |
|-------------------------|---|--------------|-------|--|--|--|--|--|--|
| Explanatory | Outcome variable: shadow economy | | | | | | | | |
| | Coefficient | t-statistics | Prob. | | | | | | |
| ΔCorruption | 0.257*** | 5.469 | 0.000 | | | | | | |
| ∆Institutional Quality | -0.799*** | -10.397 | 0.000 | | | | | | |
| ∆Growth per capita | -0.018** | -2.941 | 0.011 | | | | | | |
| ΔGov./GDP | 0.060*** | 4.799 | 0.000 | | | | | | |
| ΔPolity2 | 0.037*** | 4.394 | 0.000 | | | | | | |
| ECM(-1) | -0.791*** | -3.036 | 0.003 | | | | | | |
| Constant | -0.015 | -0.221 | 0.829 | | | | | | |
| R-squared | 0.834 | | | | | | | | |
| R-bar-squared | 0.702 | | | | | | | | |
| Durbin-Watson | 2.025 | | | | | | | | |

Source: Author's calculation. **, ***, indicate statistical significance level at 5% and 1%, respectively.

support the provision of goods and services that improve the welfare of the poor. The impact of this is that the incentive for entrepreneurs to operate in the informal sector is dampened since the formal sector can address any gaps in the provision of public goods and services which are essential for improving the lives of the poor.

Additionally, we find that government spending (as % of GDP) is another key driver of informality in Uganda in the long-run. We can observe that the coefficient on Gov/Gdp is positive and statistically significant at 1% level. Specifically, an increase in government spending as a proportion of GDP significantly increases the size of the shadow economy by 0.063 units. This is consistent with Hibbs and Piculescu (2005) who emphasize that the "grabbing hands" of the corrupt official could also potentially become the "helping hands" for entrepreneurs who see the "grabbing hands" as an "extra burden" to their business hence inducing these entrepreneurs to conceal their operations from the regulators. Similarly, Buehn and Schneider (2012b) study the determinants of the shadow economy and find evidence that high taxes induce entrepreneurs to operate underground. Our results seem to mirror this conclusion. Furthermore, the results also indicate that governance is important in influencing the size of the shadow economy in the longrun. Contrary to the literature showing that good governance reduces the size of the shadow economy (see Esaku, 2021c; Teobaldelli & Schneider, 2013), our results seem to suggest that it encourages informality especially where there is widespread corruption. Specifically, we find that an improvement in governance significantly increases the size of the informal sector by 0.026 units, statistically significant at 5% level.

As a robustness check, Menegaki (2019) advises to use DOLS (dynamic OLS) and FMOLS (fully modified OLS) to validate the robustness of the results. This is because the two econometric techniques generate asymptotically efficient coefficients and can address the issue of endogeneity and serial autocorrelation (Menegaki, 2019). Additionally, Ali et al. (2017) has shown that DOLS is important because it accommodates the mixed order of integration of variables in the cointegration framework. We report these results in Table 5 columns 4 and 6. These results are qualitatively and quantitatively similar to the ARDL results. We can observe that an increase in corruption positively and significantly increases the size of the shadow economy in the long-run. For Uganda's case, this paper finds that corruption and the size of the shadow economy are highly correlated in the long-run.

In sum, these findings confirm the expectation of this study by showing a positive relationship between corruption and the shadow economy. Subsequently, it can be concluded that an increase in corruption, government spending and polity2 do increase the size of the shadow economy in Uganda. However, economic growth and institutional quality could be viewed as effective remedies against the expansion of the shadow economy since an increase in the level of these variables lead to a decrease in the size of the shadow economy.

5.3. The short-run relationship between the shadow economy and corruption

In this section, we report the results of short-run relationship between the shadow economy and corruption. The findings are shown in Table 6, column 2. We can note that, in the short-run, there is a statistically significant relationship between the shadow economy and corruption. The above results show that corruption has a statistically significant effect on the size of the shadow economy, significant at 1% level. Specifically, the short-run results indicate that an increase in corruption significantly increases the size of the shadow economy by 0.257 units, statistically significant at 1% level, all else equal. This might indicate that a country that fails to control corruption, could actually be encouraging shadow activities in the economy. The implication of this result is that corruption significantly induces entrepreneurs to go underground in the short-run as well as the long-run. As entrepreneurs initiate operations, the business and regulatory environment is important in their consideration to either operate in the formal economy or join the informal economy. If there is any possibility for the corrupt tax official to ask for bribes in the initial stages of business registration or operations, this could be an incentive for entrepreneurs to collude with the tax official and continue operating underground as shown by Hindricks et al. (1999) and Johnson et al. (1998).

Correspondingly, this paper also finds evidence confirming the importance of institutional quality in shaping the size of the shadow economy. Specifically, we establish that an improvement in the quality of institutions significantly reduces the size of shadow economy by 0.799 units, statistically significant at 1% level, all else equal. This is consistent with the literature showing the importance of institutional quality in reducing the size of the shadow economy (see Elgin & Oztunali, 2014). Additionally, Dreher et al. (2009) present a model that captures how the interrelationship between the shadow economy, corruption and institutional quality unfold and find evidence of a negative effect of institutional quality on corruption. Their paper shows that an improvement in institutional quality negatively affects the shadow economy and reduces corruption. This implies that improvement in institutional quality undermines the

ability of corrupt officials to ask for bribes from entrepreneurs, which in turn encourages these businesses to operate in the formal economy rather than operate in the underground economy.

Additionally, the short-run results indicate that economic growth is important in mitigating the size of the shadow economy. We observe that the coefficient on growth is negative and statistically significant at 5% level. This implies that an increase in the level of economic growth significantly reduces the size of the shadow economy by 0.018 units. This is consistent with the finding in Buehn and Schneider (2012b) which shows that the level of economic development is another main determinant of informality. Furthermore, we find evidence of a short-run positive and statistically significant relationship between the size of the shadow economy and government spending. The short-run results indicate that a rise in government spending increases the size of the shadow economy by 0.060 units, statistically significant at 1% level. This indicates that high taxes do drive entrepreneurs and individual households to conceal their business operations from tax bodies hence an increase in informal sector activities, consistent with the suggestions in Buehn and Schneider (2012b). Correspondingly, the results also show evidence of a positive relationship between governance and the size of the shadow economy. Contrary to the literature, we find that an improvement in governance promotes informal sector activities. Specifically, we find that an improvement in governance significantly increases the size of shadow activities by 0.037 units, statistically significant at 1% level. Although this seems to contradict the literature, one possible explanation could be that democracy may not temper informality in the presence of widespread corruption.

Finally, we analyze how the shadow economy adjusts to any deviations from the equilibrium. This is done by examining the size and significance of the coefficient on the lagged error correction term (ECT). We can observe that the lagged coefficient of ECT is negative, between zero and one, and statistically significant at 1% level. Specifically, the ECM results indicate that the shadow economy adjusts to any deviations from long-run equilibrium at a speed of adjustment that is, 79.1%, statistically significant at 1% level.

In sum, this paper provides evidence of a positive relationship between corruption and the shadow economy in Uganda. These results seem to suggest that corruption could be partly responsible for the rise of informal sector activities in low-income countries like Uganda given that these countries have weak institutions that cannot effectively fight corruption. The practical

| Table 7. Long run e | Table 7. Long run estimates | | | | | | | | | |
|---------------------|----------------------------------|---------|-------------|---------|-------------|---------|--|--|--|--|
| Explanatory | Outcome variable: shadow economy | | | | | | | | | |
| variables | AR | DL | DC | OLS | FMOL | S | | | | |
| | Coefficient | t-stat. | Coefficient | t-stat. | Coefficient | t-stat. | | | | |
| Corruption | -0.015* | -2.129 | -0.015** | -3.050 | -0.017** | -3.526 | | | | |
| Inst. Quality | -0.100 | -0.527 | -0.100 | -0.755 | -0.040 | -0.082 | | | | |
| Growth | -0.015* | -2.355 | -0.015** | -3.374 | -0.013** | -2.839 | | | | |
| GOV./GDP | 0.057*** | 5.627 | 0.057*** | 8.061 | 0.053*** | 7.433 | | | | |
| Polity2 | -0.056* | -3.514 | -0.056*** | -5.035 | -0.056*** | -5.381 | | | | |
| Constant | 0.870 | 1.519 | 0.870* | 2.176 | 0.379 | 0.703 | | | | |
| R-Squared | 0.899 | | 0.824 | | 0.798 | | | | | |
| R-bar-Squared | 0.733 | | 0.701 | | 0.671 | | | | | |
| Durbin Watson | 2.208 | | | | | | | | | |

Source: Author's calculation. *,**, and ***, indicate statistical significance level at 10%, 5% and 1%, respectively.

| Table 8. Short run esti | Table 8. Short run estimates | | | | | | | |
|-------------------------|----------------------------------|--------------|-------|--|--|--|--|--|
| Explanatory | Outcome variable: shadow economy | | | | | | | |
| | Coefficient | t-statistics | Prob. | | | | | |
| ΔCorruption | -0.020* | -2.395 | 0.075 | | | | | |
| ∆Institutional Quality | -0.682* | -1.807 | 0.083 | | | | | |
| ∆Growth per capita | -0.022** | -4.057 | 0.015 | | | | | |
| ΔGov./GDP | 0.047** | 3.871 | 0.018 | | | | | |
| ΔPolity2 | -0.056** | -3.114 | 0.017 | | | | | |
| ECM(-1) | -0.673*** | -4.321 | 0.000 | | | | | |
| Constant | -0.273* | 2.114 | 0.062 | | | | | |
| R-squared | 0.893 | | | | | | | |
| R-bar-squared | 0.705 | | | | | | | |
| Durbin-Watson | 1.852 | | | | | | | |

Source: Author's calculation. *,**, and ***, indicate statistical significance level at 10%, 5% and 1%, respectively.

| Table 9. Model diagnostics | | | | |
|--|-------------|-------|----------------------|-------|
| Equation (1) | | | | |
| Normality test | | | Jarque-Bera | 0.428 |
| | | | Probability | 0.807 |
| Breusch-Godfrey serial correlation LM Test | F-statistic | 0.681 | Prob.F(2,14) | 0.522 |
| | Obs*R-sq. | 2.039 | Prob. Chi-Squared(2) | 0.361 |
| Heteroskedasticity Test Breausch-Pagan- Godfrey | F-statistic | 1.785 | Prob.F(6,6) | 0.166 |
| | Obs*R-sq. | 9.223 | Prob. Chi-Squared(6) | 0.161 |
| Equation (2): Alternative specification | | | | |
| Normality test | | | Jarque-Bera | 1.204 |
| | | | Probability | 0.548 |
| Breusch-Godfrey serial correlation LM Test | F-statistic | 0.222 | Prob.F(2,2) | 0.818 |
| | Obs*R-sq. | 2.182 | Prob. Chi-Squared(2) | 0.336 |
| Heteroskedasticity Test Breausch-Pagan- | F-statistic | 1.044 | Prob.F(7,4) | 0.514 |
| Godfrey | Obs*R-sq. | 7.756 | Prob. Chi-Squared(7) | 0.355 |

Source: Author's calculations.

implication of the findings in this paper is that tackling informality requires carrying political and institutional reforms to empower institutions to tackle corruption. Secondly, policy effort be directed at strengthening enforce of existing laws to be able to detect, investigate and prosecute corrupt officials and individual who promote corrupt practices. This policy framework should include a policy package that makes it harder or nearly impossible for the corrupt tax official to ask for bribes from entrepreneurs.

5.4. Alternative specification

To test the sensitivity of our results, this paper uses an alternative specification of eq. (1), where corruption-based indices are used instead of index of corruption. We report the results of the empirical analysis for the long-run relationship in Table 7 and short-run in Table 8. In both of these tables, we can observe that corruption seems to reduce the size of the shadow economy activities instead of increasing them. This finding seems to agree with Dreher and Schneider (2010) study

| Table 10. Lo | Table 10. Long-run granger causality results (Wald F-statistics) | | | | | | | | | |
|--------------|--|-----------------|------------------|------------------|------------------|-----------------|--|--|--|--|
| | shadow E | Corruption | Ins.qualty | Growth | Gov/GDP | Pol2 | | | | |
| Shadow E | - | 9.689(0.005)*** | * 7.779(0.004)** | * 6.453(0.008)** | * 4.341(0.015)** | 1.306(0.298) | | | | |
| Corruption | 3.347(0.061)* | - | 4.511(0.028) | 1.008(0.387) | 3.500(0.055)* | 9.905(0.002)*** | | | | |
| Ins.qualty | 1.793(0.198) | 5.740(0.012)** | - | 1.423(0.269) | 3.351(0.061)* | 0.424(0.661) | | | | |
| Growth | 2.559(0.107) | 0.020(0.980) | 0.474(0.631) | - | 0.335(0.720) | 0.603(0.559) | | | | |
| Gov/GDP | 0.989(0.392) | 0.791(0.470) | 0.845(0.447) | 2.796(0.088)* | - | 0.881(0.433) | | | | |
| Pol2 | 11.597(0.001)* | *1.558(0.241) | 6.709(0.007)** | * 5.071(0.019)* | 6.794(0.007)** | * _ | | | | |

Source: Author's calculation. *,**,***, indicate statistical significance at 10%, 5% and 1% respectively.

which reveals that using perception-based indices of corruption is controversial and sometimes results into no robust relationship. In line with the above, one can argue that perception-based indices of corruption may indeed result into some debatable results. In what follows, we argue that the relationship between the size of the shadow economy and corruption is sensitive to the corruption indices used. The use of perception-based indices of corruption may, in some cases lead to either negative relationship or no significant relationship between the size of the shadow economy and corruption in both the long-run and short-run (see Dreher & Schneider, 2010).

5.5. Diagnostic tests

To confirm the reliability of these results we perform a number of diagnostic tests, such as residual and stability diagnostics. We test whether these results are driven by biases that could affect their reliability. We carry out Normality, Breusch-Godfrey serial correlation LM and Heteroskedasticity; Breusch-Pagan-Godfrey (BPG) tests and report the results of these tests in Table 9. The results of these tests show no evidence of any biases in the empirical estimation. This paper also reports the plots of cumulative sum of recursive residuals (CUSUM) and the plots of cumulative sum of squares of recursive residuals (CUSUMQ). We present these plots in Figures 1-4. As shown in these figures, the residual plots are tightly bound within the boundaries at a significance level of 5%. These plots provide additional evidence on the stability of the estimated model. As shown in these figures, the residual plots confirm the reliability of these findings since they show evidence of stability in the coefficients of the estimated ARDL models. In what follows, we summarize by arguing that these results show evidence of a positive and statistically significant relationship between the size of the shadow economy and corruption.



Figure 1. Plot of cumulative sum of recursive residuals for ARDL model (1).

Figure 2. Plot of cumulative sum of squares of recursive residuals for ARDL model (1).







5.6. Granger causality test

5.6.1. Long-run granger causality results

In this section, we conduct the pairwise granger causality test to ascertain whether there is a causal relationship among the variables. Our main interest is to determine whether the relationship between corruption and the size of the shadow economy is bidirectional in both the long-run and short-run. Table 10 reports the long-run results for the granger causality test. We can observe that, in the long-run, there is a causal relationship between the size of the shadow economy and corruption, where causality runs from the shadow economy to corruption and from corruption to the shadow economy. Thus this study establishes a long-run bidirectional relationship between corruption and the size of the shadow economy. This is consistent with the suggestion of Buehn and Schneider (2012a) which emphasizes a bidirectional relationship between the size of the shadow economy and corruption. Additionally, we also note a causal relationship between institutional quality and the size of the shadow economy, where causality runs from institutional quality to the size of the shadow economy.

Figure 4. Plot of cumulative sum of squares of recursive residuals for ARDL model (2).



This could suggest that an improvement in the quality of institutions influences the size of the shadow economy (see Elgin & Oztunali, 2014). Similarly, we also observe a causal relationship between growth and the size of the shadow economy; where causality runs from growth to the shadow economy. This would agree with the literature that emphasizes the importance of the level of development in influencing the size of the shadow economy (see Esaku, 2021b). Similarly, this paper finds evidence of a causal relationship between government spending and the size of the shadow economy, running from government spending to shadow economy. The evidence shows that an increase in government spending increases the size of the shadow economy, consistent with the literature (see Buehn & Schneider, 2012b; Esaku, 2021e). From row 2, columns 6 and 7, there is evidence of a causal relationship between government spending and corruption, and between governance and corruption in the long-run. The causal results show that an increase in government spending to corruption. Additionally, these results also indicate that more democracy (governance) causes corruption. This result seems to agree with the study in Jetter et al. (2015) which established that democracy worsens corruption especially in countries where the income level is low.

Turning to row 3, columns 3 and 6; we find a causal relationship running from corruption to institutional quality. Additionally, there is also causality running from government spending to institutional quality. This appears to suggest that a rise in corruption leads to the creation of institutions aimed at combating it. Furthermore, we also observe a causal relationship between growth and government spending. The results suggest that more growth significantly increases government spending due to the possible rise in revenue. Finally, we turn to row 7, columns 2, 4, 5 and 6. We observe that an increase in shadow activities influences the governance system, with causality running from shadow economy to governance. Additionally, improvement in institutional quality significantly causes good governance. Furthermore, there is evidence that improvement in economic growth and increase in government spending causes good governance.

Taken together, the pairwise causality results of the long-run relationship between corruption and the size of the shadow economy indicate a causal long-run relationship between corruption and the size of the shadow economy which is bidirectional.

5.6.2. Short-run granger causality results

In this section, we report the pair wise granger causality results for the short-run. The results are reported in Table 11. In column 2, and 3; rows, 2, and 3, there is evidence of a causal relationship

| Table 11. Short-run granger causality results (Wald F-statistics) | | | | | | |
|---|----------------|----------------|--------------|----------------|--------------|------------------|
| | ∆shadow E | ΔCorruption | ∆Ins.qualty | ∆Growth | ∆Gov/GDP | ΔPol2 |
| ∆Shadow E | - | 4.233(0.024)** | 2.233(0.142) | 1.614(0.230) | 1.034(0.378) | 0.515(0.607) |
| ∆Corruption | 6.123(0.011)** | - | 0.154(0.859) | 5.042(0.021)** | 1.246(0.316) | 12.357(0.000)*** |
| ∆Ins.qualty | 4.096(0.038)** | 0.498(0.618) | - | 2.085(0.159) | 0.106(0.900) | 4.893(0.023)** |
| ∆Growth | 5.624(0.014)** | 0.083(0.921) | 2.693(0.100) | - | 0.713(0.505) | 0.295(0.749) |
| ∆Gov/GDP | 0.166(0.848) | 0.060(0.921) | 0.252(0.781) | 1.689(0.214) | - | 0.461(0.639) |
| ΔPol2 | 0.333(0.722) | 0.023(0.977) | 0.820(0.459) | 0.755(0.487) | 0.461(0.639) | - |

Source: Author's calculation. **, ***, indicate statistical significance at 5% and 1% respectively.

between the shadow economy and corruption. We can observe that shadow activities cause corruption, but also corruption causes shadow activities. Consequently, this paper establishes a short-run bidirectional causal relationship between shadow economy and corruption, consistent with the study of Buehn and Schneider (2012b). The results also indicate a causal relationship between shadow economy and institutional quality, with causality running from shadow economy to institutional quality. Similar results can be observed between shadow economy and growth, with causality running from shadow economy to growth. This could be suggesting that a large shadow economy undermines economic growth (see Esaku, 2021b). Correspondingly, the short-run results also indicate a causal relationship between the level of economic growth and corruption, with causality running from growth to corruption. This would imply that more growth creates corruption in the short-run (see Brown & Shackman, 2007). Similarly, there is also evidence of a short-run causal relationship between governance and corruption, with causality running from governance to corruption. This could suggest that bad governance promotes corruption as has been shown by Quah (2009). Additionally, we find evidence of a causal relationship between governance and institutional quality, with causality running from governance to institutional quality.

In sum, this paper establishes a short-run as well as long-run causal relationship between corruption and the size of the shadow economy in Uganda. This is a significant finding given that these factors (corruption and shadow activities) have been in the rise in the country. At the policy level, addressing the rise in informality requires reforming the political and institutional system to detect, investigate and prosecute cases of corruption in the corruption.

6. Conclusion

In this paper, we examine the relationship between the size of the shadow economy and corruption in a country-level analysis using ARDL modeling approach. We use annual time series data covering the period from 1984 to 2008, drawn from several data sources. The results of the empirical analysis on the relationship between the size of the shadow economy and corruption reveal a positive and statistically significant relationship, in both the long-run and short-run. Furthermore, the ARDL results that establish a long-run and short-run relationship between corruption and the size of the shadow economy are also backed by the granger causality results. These findings suggest that, a bidirectional causal relationship between corruption and the size of the shadow economy. Similarly, increase in corruption leads to an increase in the size of the shadow economy. Similarly, increase in informality also leads to corruption. These results line up well with the view that corruption and informality are complements, and are detrimental to the growth of economies. This implies that corruption and shadow activities distort the proper functioning of the market mechanism and undermine effort to improve the welfare of the poor. One can argue that the positive relationship between corruption and the size of the

shadow economy is expected, especially in low-income countries such as Uganda where the two variables are on the rise.

Overall, this paper provides evidence that seems to suggest that corruption could be responsible for the rise of informal sector activities in low-income countries like Uganda given the weak institutional framework that is ineffective in fighting corruption. Additionally, a large size of the shadow economy in Uganda could also be the result of corruption.

Given the complementary relationship between corruption and the size of the shadow economy, addressing widespread informality in the country would require; first, reforming the political system to have the political will to tackle political corruption and go after politicians who use their influence and power to circumvent institutions. Second, carrying out institutional reforms to address political patronage and influence peddling would go a long way into addressing systemic corruption which in turn could help mitigate the spread of informal sector activities. Third, strengthening the enforcement of existing laws to identify and punish culpable public officials who use their offices for private gain would also address the level of informality in the country.

We acknowledge that this study faces two main limitations. First this study uses data on the size of the shadow economy that cover 25 years. We suggest future studies may benefit from using historical data on size of shadow economy that cover more years. Second, data on the size of the shadow economy is aggregated at the national economy level making it difficult to explore how corruption affects the informal activities in various sector of the economy. Disaggregating the data on the size of the shadow economy would facilitate this analysis in future. Finally, future research could focus on developing a theoretical framework that explains the various channels through which corruption influences the size of the shadow economy

Funding

The author received no direct funding for this research.

Author details

Stephen Esaku¹ E-mail: esaku_stephen@yahoo.com ORCID ID: http://orcid.org/0000-0002-2587-4092

¹ Department of Economics, Statistics Kyambogo University, Soroti city, Uganda.

Citation information

Cite this article as: Does corruption contribute to the rise of the shadow economy? Empirical evidence from Uganda, Stephen Esaku, *Cogent Economics & Finance* (2021), 9: 1932246.

Notes

- 1. We follow Dreher and Schneider (2010), and define corruption as "the abuse of pubic power for private gain."
- 2. In this paper, we follow Schneider (2005) and define the shadow economy as "all market-based legal production of goods and services that are intentionally hidden from regulators." The use of the word shadow economy also means the informal economy (sector) and underground economy. We use these words interchangeably.

References

Ali, W., Abdullah, A., & Azam, M. (2017). Revisiting environmental Kuznets curve hypothesis for for Malaysia: Fresh evidence from ARDL bounds testing approach. Renewable and Sustainable. *Energy Reviews*, 77, 990–1000. https://doi.org/10.1016/j.rser.2016.11.236

- Blackburn, K., Bose, N., & Capasso, S. (2012). Tax evasion, the underground economy and financial development. Journa of Economic Behaviour and Organization, 83(1), 243–253. https://doi.org/10. 1016/j.jebo.2012.05.019
- Brown, S. F., & Shackman, J. (2007). Corruption and related socioeconomic factors: A time series study. *Kyklos*, 60(3), 319–347. https://doi.org/10.1111/kykl. 2007.60.issue-3
- Buehn, A., & Schneider, F. (2012a). Corruption and the shadow economy: Like oil and vinegar, like water and fire? International Tax and Public Finance, 19(1), 172–194. https://doi.org/10.1007/s10797-011-9175-y
- Buehn, A., & Schneider, F. (2012b). Shadow economies around the world: Novel insights, accepted knowledge, and new estimates. *International Tax and Public Finance*, 19(1), 139–171. https://doi.org/10. 1007/s10797-011-9187-7
- Capasso, S., & Jappelli, T. (2013). Financial development and the underground economy. *Journal of Development Economics*, 101, 167–178. https://doi. org/10.1016/j.jdeveco.2012.10.005
- Choi, J. P., & Thum, M. (2005). Corruption and the shadow economy. International Economic Review, 46(3), 817–836. https://doi.org/10.1111/j.1468-2354.2005. 00347.x
- De Soto, H. (1989). The other path: The Economic answer to terrorism. Harper and Row.
- Dreher, A., Kotsogiannis, C., & McCorriston, S. (2009). How do institutions affect corruption and the shadow economy? International Tax and Public Finance, 16(6), 773–796. https://doi.org/10.1007/s10797-008-9089-5
- Dreher, A., & Schneider, F. (2010). Corruption and the shadow economy: An empirical analysis. *Public*

Choice, 144((1-2)), 215-238. https://doi.org/10.1007/ s11127-009-9513-0

- Elgin, C., & Oztunali, O. (2012). Shadow economies around the world: Model based estimates. *Working Paper no.* 2012/05, Bogazici University, Department of Economics, Istanbul
- Elgin, C., & Oztunali, O. (2014). Institutions, informal economy, and economic development. *Emerging Markets Finance and Trade*, 50(4), 145–162. https:// doi.org/10.2753/REE1540-496X500409

Enders, W. (2004). Applied econometric time series (2nd ed ed.). John Wiley & Sons.

- Esaku, S. (2021a). Does the shadow economy increase income inequality in the short- and long-run? Empirical evidence from Uganda. *Cogent Economics* & Finnce, 9(1), 1–21. https://doi.org/10.1080/ 23322039.2021.1912896
- Esaku, S. (2021b). Is informality a barrier to economic growth in Ugnda? Empirical analysis. *Development Studies Research*, 8(1), 109–121. https://doi.org/10. 1080/21665095.2021.1919167
- Esaku, S. (2021c). Political instability and informality in Uganda: An empirical analysis. *The European Journal* of Applied Economics, Forthcoming. pp. 1-16. https:// aseestant.ceon.rs/index.php/sjas/authorDashboard/ submission/31159
- Esaku, S. (2021e). The long- and short-run relationship between the shadow economy and trade openness in Uganda. *Cogent Economics and Finance, Forthcoming.* https://doi.org/10.1080/23322039. 2021.1930886
- Esaku, S., & Watson, D. (2020b). Job creation, job destruction and reallocation in Sub-Saharan Africa: Firm-level evidence from Kenyan manufacturing sector. *Cogent Economics & Finance*, 8(1), 1–23. https://doi.org/10.1080/2332039.2020.1782113
- Friedman, F., Johnson, S., Kaufmann, D., & Zoido-Lobaton, P. (2000). Dodging the grabbing hand: The determinants of unofficial activity in 69 countries. *Journal of Public Economics*, 76(3), 459–493. https:// doi.org/10.1016/S0047-2727(99)00093-6
- Goel, R. K., & Saunoris, J. W. (2014). 'Global corruption and the shadow economy: Spatial aspects. *Public Choice*, 161(1–2), 119–139. https://doi.org/10.1007/s11127-013-0135-1
- Hibbs, D. A., & Piculescu, V. (2005). Institutions, corruption and tax evasion in the unofficial economy. Department of Economics.
- Hindricks, J., Murthoo, A., & Keen, M. (1999). Corruption, extortion and evasion. *Journal of Public Economics*, 74(3), 395–430. https://doi.org/10.1016/S0047-2727(99)00030-4
- International Country Risk Guide (ICGR) Researchers Dataset. (2017). Political Risk Services (PRS). Retrieved https://doi.org/10.7910/DVN/4YHTPU, Harvard Dataverse, VI
- Jetter, M. A, Alejandra M. H, and Andrés Ramírez, (2015). The Effect of Democracy on Corruption: Income is Key. World Development, 74(C), pp. 286–304. https:// doi.org/10.1016/j.worlddev.2015.05.016
- Johnson, S., Kaufmann, D., Shleifer, A., Goldman, M. I., & Weitzman, M. L. (1997). The unofficial economy in transition. Brookings Papers on Economic Activity, 2 (2), 159–221. https://doi.org/10.2307/2534688

- Johnson, S., Kaufmann, D., & Zoido-Lobaton, P. (1998). Regulatory discretion and the unofficial economy. American Economic Review, 88(2), 387-392. https://citeseerx.ist.psu.edu/viewdoc/ download?
- doi=10.1.1.194.5747&rep=rep1&type=pdf Knack, S., & Keefer, P. (1995). Institutions and economic performance: Cross-country tests using alternative institutional measures. *Economics and Politics*, 7(3), 207–227. https://doi.org/10.1111/j.1468-0343.1995. tb00111.x
- Luong, T. T. H., Ngoyen, T. M., & Nguyen, T. A. N. (2020). Rule of law, economic growth and shadow economy in transition countries. *Journal of Asian Finance, Economics and Business*, 7(4), 145–154. https://doi. org/10.13106/jafeb.2020.vol7.no4.145
- Marshall, M. G., Gurr, T. R. & Jaggers, K. (2018). Polity5 dataset version 2018: Political regime characteristics and transitions, 1800-2018. Center for Systemic Peace. Retrieved http://www.systemicpeace.org/ inscrdata.html
- Menegaki, A. N. (2019). The ARDL method in the energy-growth nexus field; best implementation strategies. *Economies*, 7(4), 1–16. https://doi.org/10. 3390/economies7040105
- Mugoda, S., Esaku, S., Nakimu, R. K., Bbaale, E., & Read, R. (2020). The portrait of uganda's informal sector: What main obstacles do the sector face? *Cogent Economics and Finance*, 8(1), 1–29. https://doi.org/10. 1080/23322039.2020.1843255
- Pesaran, H., Shin, Y., & Smith, R. J. (2001). Bounds testing approaches to the analysis of level relationships. *Journal of Applied Econometrics*, 16(3), 289–326. https://doi.org/10.1002/()1099-1255
- Quah, J. S. T. (2009). Governance and corruption: Exploring the connection. American Journal of Chinese Studies, 16(2), 119–137.
- Schneider, F. (2005). Shadow economies around the world: What do we really know? European Journal of Political Economy, 21(3), 598–642. https://doi.org/10. 1016/j.ejpoleco.2004.10.002
- Schneider, F. (2007). Shadow economies and corruption all over the world: New estimates for 145 countries. *Economics E-jornal*, 1, 2007–2009.
- Schneider, F., & Enste, D. H. (2000). Shadow economies: Size, causes, and consequences. Journal of Economic Literature, 38(1), 77–114. https://doi.org/10.1257/jel. 38.1.77
- Tang, T. C. (2004). A reassessment of aggregate import demand function in the Asean-5: A cointegration analysis. The International Trade Journal, 18(3), 239–268. https://doi.org/10.1080/ 08853900490478131
- Teobaldelli, D., & Schneider, F. (2013). The influence of direct democracy on the shadow economy. *Public Choice*, 157(3–4), 543–567. https://doi.org/10.1007/ s11127-013-0098-2
- Transparency International. (2019). Transparency International, Berlin Germany. Retrieved July 20, 2020, from https://www.transparency.org/en/cpi/2019/results
- World Bank, Africa Development indicators 2014 computer file. Retrieved January 13, 2020, from https://datacata log.worldbank.org/dataset/africa-developmentindicators



© 2021 The Author(s). This open access article is distributed under a Creative Commons Attribution (CC-BY) 4.0 license. You are free to:

Share — copy and redistribute the material in any medium or format.

Adapt — remix, transform, and build upon the material for any purpose, even commercially.

The licensor cannot revoke these freedoms as long as you follow the license terms.

•

Under the following terms: Attribution — You must give appropriate credit, provide a link to the license, and indicate if changes were made. You may do so in any reasonable manner, but not in any way that suggests the licensor endorses you or your use. No additional restrictions

You may not apply legal terms or technological measures that legally restrict others from doing anything the license permits.

Cogent Economics & Finance (ISSN: 2332-2039) is published by Cogent OA, part of Taylor & Francis Group. Publishing with Cogent OA ensures:

- Immediate, universal access to your article on publication
- High visibility and discoverability via the Cogent OA website as well as Taylor & Francis Online
- Download and citation statistics for your article
- Rapid online publication
- Input from, and dialog with, expert editors and editorial boards
- Retention of full copyright of your article
- Guaranteed legacy preservation of your article
- Discounts and waivers for authors in developing regions

Submit your manuscript to a Cogent OA journal at www.CogentOA.com